Author’s response to reviews

Title: Predicting Risk of Hospital and Emergency Department Use for Home Care Elderly Persons through a Secondary Analysis of Cross-National Data

Authors:

John N Morris (jnm@hsl.harvard.edu)
Elizabeth P Howard (e.howard@neu.edu)
Knight Steel (knightrsteel@yahoo.com)
Robert Schreiber (rschreiber@hsl.harvard.edu)
Brant E Fries (bfries@umich.harvard.edu)
Lewis A Lipsitz (lipsitz@hsl.harvard.edu)
Beryl Goldman (bgoldman@kendaloutreach.org)

Version: 4 Date: 26 September 2014

Author’s response to reviews:

Abstract reduced from 386 words to 350 words in accordance with the guidelines.